Author’s response to reviews

Title: Stakeholder perceptions of a nurse led walk-in centre.

Authors:

Rhian M Parker (rhian.parker@anu.edu.au)
Jane L Desborough (jane.desborough@anu.edu.au)
Laura E Forrest (laura.forrest@anu.edu.au)

Version: 5 Date: 23 October 2012

Author’s response to reviews: see over
Thank you for the updated review of this paper. The reviewer acknowledges that the main concerns noted about the first submission have been addressed. In response the reviewer’s comment about relevance, the authors believe that understanding stakeholder perceptions is crucial when novel health services are developed. Stakeholders have the potential to support or reject such services and this has implications for the development of policy and the way in which new service models are planned and negotiated.

Thank you,
Rhian Parker